Author's response to reviews

Title: Dietary patterns associated with fall related fracture in elderly Japanese: a population based prospective study

Authors:

Yasutake Monma (monmayasutake@mail.tains.tohoku.ac.jp)
Kaijun Niu (ggg@mail.tains.tohoku.ac.jp)
Koh Iwasaki (QFG03604@nifty.com)
Naoki Tomita (n.tomita@ab.cyberhome.ne.jp)
Naoki Nakaya (nakaya-thk@umin.ac.jp)
Atsushi Hozawa (hozawa-thk@umin.ac.jp)
Shinichi Kuriyama (kuriyama-thk@umin.ac.jp)
Takashi Takeda (take@m.tains.tohoku.ac.jp)
Nobuo Yaegashi (yaegashi@mail.tains.tohoku.ac.jp)
Satoru Ebihara (sebihara@idac.tohoku.ac.jp)
Hiroyuki Arai (harai@idac.tohoku.ac.jp)
Ryoichi Nagatomi (nagatomi@m.tains.tohoku.ac.jp)
Ichiro Tsuji (tsuji1@mail.tains.tohoku.ac.jp)

Version: 2 Date: 3 December 2009

Author’s response to reviews: see over
Dear Dr Norton:

Please find a revised version of our manuscript titled above. We would like to thank you and the reviewers for your constructive comments and have revised our manuscript accordingly. According to your suggestion, we changed the manuscript, figure and table. We revised our manuscript as described in a point-by-point response to reviewers. Changes to the text of this manuscript are underlined. This work has not been published elsewhere, nor is it being considered for publication elsewhere. We guarantee the accuracy of our references. We believe that the manuscript has been greatly improved as a result of your suggestions, and we hope that it will be found to be acceptable for publication in *BMC Geriatrics*.

Thank you very much for your kind consideration.

Sincerely,  
Koh Iwasaki

---

Melissa Norton, MD  
Editor-in-Chief, *BMC Geriatrics*  
United Kingdom

Koh Iwasaki, MD, PhD.  
Associate Professor,  
Center for Asian Traditional Medicine,  
Graduate school of medicine, Tohoku University  
Seiryo-machi, Aoba Ward,  
Sendai city, Miyagi pref., Japan  
Phone: +81-22-717-7185, FAX: +81-22-717-7186  
Email QFG03604@nifty.com

MS: 7331091202862740 Dietary patterns associated with fall related fracture in elderly  
Japanese: a population based prospective study by Y Monma et al.